EO-1 Weekly Status Week of December 24 – December 30, 2009

Day of Year 358 - 364
Mission Day 3330 - 3336
Earth Observing-One (EO-1) General

Scheduled 131 Science Data Collection Events (DCEs) this week.
INSTRUMENTS

All instruments operated nominally this week 

Performed instrument calibration

· Conducted ALI internal calibration (Type II) on December 28 at 00:00z

EO-1 Spacecraft Subsystems

Command and Data Handling (C&DH)

Continued to experience problems playing back engineering data from the Solid-State-Recorders.  Engineering data is being received during real-time contacts and for all science imaging events. 

Technology Activities

In addition to providing ongoing science data collection, the EO-1 extended mission supports on-orbit testbed activities for advanced technology and hyperspectral research.  The status of various validation efforts is contained in the following paragraphs.

SensorWeb & Virtual Observatory Demonstrations
No activities to report for this week’s period.
Autonomous Sciencecraft Experiment (ASE)

The ASE controlled EO-1 all week. 

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Continued nominal spacecraft state of health

· Continued mitigation of low voltage values for recent modified lunar calibrations

· Received notification that mission scientist would prefer to temporarily suspend modified lunar calibrations while alterations are developed

· Conduct only nominal calibration in January

· Plan to begin study of potential changes to modified calibration sequence

· Continued analysis of recent increase in frequency of some instrument temperature limit violations—most notably the temperature of the ALI power module

· Analyzed end of year data for 2009 (after day 335) and discovered temperatures decreasing below typical seasonal levels

· Observed decreased number of images being scheduled per day 

· Beginning to quantify effect of images per day (and image types) on temperatures in order to develop operational constraints

· Continued study of attitude control system position error values

· Found errors about y-axis to be more noisy than other axes

· Observed decrease in driver voltage for B gyro in IRU during mission lifetime that has been more steep than other two gyros

· Performing study of lifetime trend to quantify rate of change in driver voltages

Mission Planning

· Continued migration from Command Management System (CMS) software to ASIST SCP (Stored Command Processor) module

· Continued functional testing

· Discovered difference between SCP and CMS regarding timestamps for automatic buffer switch commands

· Found CMS places buffer switch at same time as last command (typically 23:59:30z)

· Found SCP always places buffer switch at 23:59:30z

· Plan to discuss issue with ASIST developers to mitigate difference
Flight Dynamics
· Tracking debris object 81716

· Received first notification of object having non-zero probability of conjunction on Dec. 29, with Pc = 0.01%

· Received update on morning of Dec 30 stating PC was now 0.02%

· Observed in-track and cross-track miss distances trending in the direction of zero

· Received notice that the time of closest approach will be 10:57:54z on Dec 31

· Studying conjunction event in collaboration with GSFC CA group

· Communicating with project management to determine if necessary to conduct a debris avoidance maneuver

· Upon considering the predicted low probability of conjunction, the uncertainty of the debris object orbit due to the infrequent tracking updates, and the late notice given of the conjunction probability that precluded double checking an avoidance maneuver commands, a decision was reached to not perform an avoidance maneuver.   
Trending

· Ingesting data into ITPS database after computer recovery   
System Administration
· Received error from mission planning computer that it is unable to print

· Awaiting return of system administrator after holidays

GROUND AND SPACE NETWORK

Station Downtimes

· Received notification that HGS is still red, effective starting September 7, and ETRO is unknown.
Operational Discrepancies
· Observed no operational discrepancies
UPCOMING EVENTS

· Conduct nominal lunar calibration on January 1
· Cancel associated modified lunar calibration 

· In January 2010, institute ASE software patch onboard spacecraft to correct for the leap second issue, the WARP Mongoose V single bit error issue, and the temperature log size issue. 

Imagery Status

Scenes and Engineering Cals planned for week of December 24 – December 30, 2009             131
Total scenes and engineering calibrations planned for entire mission – approximately 50,587
Total Scenes:  ALI scenes in the Level 0 archive              46,203             (as of December 30, 2009)

                         Hyperion scenes in the Level 0 archive    45,955             
Publications and Presentations Status 

404 publications 

284 external presentations

53 articles and press releases
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